Author’s response to reviews

Title: Association of ABO Polymorphisms and Pancreatic Cancer/ Cardiocerebrovascular Disease: a Meta-analysis

Authors:

Lian Li (lilian@tmu.edu.cn)
Yanxia Li (liyanxia0525@163.com)
Luyang Liu (iuluyang0628@163.com)
Yubei Huang (yubei_huang@163.com)
Hong Zheng (zhengh64@aliyun.com)

Version: 3 Date: 11 Feb 2020

Author’s response to reviews:

1. We note a change in the order of authors, since the original submission of the manuscript.

In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts. We would therefore need to ask that all authors complete our change of authorship form, which can be downloaded from http://www.biomedcentral.com/submissions/editorial-policies#authorship.

Please read the instructions in the form carefully, and return the form within 14 days with all signatures (including from any newly proposed/removed co-author(s)) to BMC Series Editorial@biomedcentral.com. Please also see our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Response: We apologize for the mistake. The mistake was happened in the submission step when submitting the manuscript. The order of authors has not changed in the title page of manuscript during the period of revision. We will send the signed form by the email soon.

2. Please only identify only one corresponding author on the title page of your manuscript, ensuring they are the same in the title page as in the submission system. Please also ensure that the email addresses are the same.

Response: There are two corresponding authors (Lian Li and Hong Zheng) in this manuscript. Therefore, we listed two corresponding author in the title page of main manuscript. However, we found that we could check only one corresponding author in submission system.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions.

Response: We have uploaded a clean version of main text, tables, figures, and additional files in this version of manuscript.

4. Please ensure all figures, tables and additional files are cited in the text, as they may be removed by the production department. Please delete any additional files not intended for publication.

Response: We have checked all the figures, tables and additional files.